

Dear Reviewers:

Thank you for your letter and for the reviewers' comments concerning our manuscript No.87054 entitled "Management of acute carbamazepine poisoning: A narrative review". Those comments are all valuable and very helpful for revising and improving our paper, as well as the important guiding significance to our research. We have carefully studied these comments and have made correction which we hope to meet with approval. The manuscript has also been double-checked, and the grammar, spelling, logic, and sentence structure errors we found have been corrected and the revised manuscript had got the Grade A. The paper was coauthored by Luan Wang, Yang Wang, Ruo-Ying Zhang, Yao Wang, Wei Liang, Tie-Gang Li, and all author information has not changed. We hope our revised manuscript can be accepted for publication. The main corrections in the paper and the responds to the reviewer's comments are as following:

Reviewer #1:

Specific Comments to Authors: This review was provided currently available information on acute CBZ poisoning, including its management, by describing and summarizing various therapeutic methods for its treatment according to previously published studies. Several treatment methods for CBZ poisoning were briefly introduced, their advantages and disadvantages were analyzed and compared, and suggestions for the clinical treatment of CBZ poisoning were provided. The text is well written with correct and clear English, using a structured methodology, and a rich and relatively recent bibliography. The title reflects the main subject very well, and the abstract summarized and reflected the work done by the authors. The manuscript interpreted the findings adequately and appropriately, emphasizing and highlighting the key points clearly and logically. I recommend accepting this review for publication after a minor language editing.

Thank for your comments and suggestion. We are sorry for our English writing. According to your suggestion, the whole manuscript has been thoroughly polished by the filipodia group to eliminate grammatical, spelling, logical and sentence structure mistakes in our manuscript. We hope it can meet the journal's standard. Thanks so much for your useful comments. The certificate of English editing was showed below.



CONFIDENTIAL
September 20, 2023

Certificate Service Confirmation

To Whom It May Concern,

Filipodia provided comprehensive editing services for **Revised Manuscript NO: 87054 (Management of acute carbamazepine poisoning: A narrative review)** by **Wang L *et al.***, which is under consideration for publication in your journal. The edit has achieved Grade A: priority publishing; no language polishing required after editing.

Should you require any additional information, please do not hesitate to contact me.

Jennifer C van Velkinburgh, PhD
President and Chief Editor & Writer,
Filipodia Publishing, LLC

Email: bpg@filipodia.com

Reviewer #2:

Specific Comments to Authors: This is a timely topic discussed by the authors. Specific antidotes are unavailable for acute CBZ poisoning due to lack of standard management protocols. Ways to address such an issue is extensively needed. The authors have provided a decent review on the topic. They describe and summarize the detailed clinical evidence on the treatment and management of CBZ poisoning, including gastric lavage and activated carbon therapy, hemoperfusion, hemodialysis, continuous renal replacement therapy, plasmapheresis and Lipid resuscitation therapy. Also, they retrospective summarize the results of CBZ poisoning-related research, and provides the best treatment plan for CBZ poisoning. Overall, the work is well-rationed,

and the relatively complete. As a review article, this paper would be helpful for clinicians to manage patients suffering from CBZintoxication effectively. Overall, I did find the manuscript is informative and worthy for the publication in this journal. I have a minor comment: The source of Table 1 needs to be specified in the comments.

Thank you for your helpful advice. At your suggestion, we have added supplement 1, which is the source of these 67 case reports. However, when i reviewing the results of the table1, due to my carelessness, I found that there was something wrong with the age in Table 1. The mean and median of age were mistaken, and the average was used instead of the median, which has been corrected in the Table1 . I'm really sorry. I hope you'll forgive me and thanks so much for your useful suggestions.

Sincerely,

Luan Wang

Department of Emergency Medicine, Shengjing Hospital of China Medical University,
Shenyang, China

36 Sanhao Street, Heping District, Shenyang City, Liaoning Province, China

08618940258821

wangluan@ sj-hospital.org